Author’s response to reviews

Title: Traditional Chinese Medicine Use Is Associated with Lower End-Stage Renal Disease and Mortality Rates among Patients with Diabetic Nephropathy: A Population-Based Cohort Study

Authors:

Hsing-Yu Chen (b8705016@gmail.com)
Heng-Chih Pan (u8801006@gmail.com)
Yung-Chang Chen (cyc2356@gmail.com)
Yu-Chun Chen (yuchn.chen@googlemail.com)
Yi-Hsuan Lin (cynthiajobo@gmail.com)
Sien-Hung Yang (dryang@adm.cgmh.org.tw)
Jiun-Liang Chen (a12015@adm.cgmh.org.tw)
Hau-Tieng Wu (hauwu@math.duke.edu)

Version: 1 Date: 16 Dec 2017

Author’s response to reviews:

Dear Editor,

All required correction were done accordingly, including email address list for all authors, and the explanation to the waived consent since that it was unable to trace back subjects’ real identities. The author list was confirmed as well Finally, the "supplementary files" section was added as well. Thank you for your consideration of our work.

Best regards,

Hau-Tieng Wu